

February 11th, 2013

Dear Editor.

Please find enclosed the edited manuscript in Word format (file name: 2006-review.docx).

Title: Evaluation of hepatic cystic lesions

Authors: Marten A Lantinga, Tom JG Gevers, Joost PH Drenth

Name of Journal: World Journal of Gastroenterology

**ESPS Manuscript NO: 2006** 

The manuscript has been improved according to the suggestions of reviewers.

- 1. The format has been updated.
- 2. The core tip was added to the manuscript.
- 3. Peer-reviewers 00054465, 00058405 and 00053684 made no suggestions for improvement.
- 4. References and typesetting were corrected.
- 5. Language evaluation. We believe our manuscript has reached grade A after revision. The manuscript was proofread by our academic proofreading service.

Thank you again for publishing our manuscript in the World Journal of Gastroenterology.

Sincerely yours,

Joost P.H. Drenth, MD PhD, Professor of Gastroenterology and Hepatology

Department of Gastroenterology and Hepatology

Radboud University Nijmegen Medical Centre

P.O. Box 9101, code 455

6500 HB Nijmegen, the Netherlands

Tel: +31 24 3614760; Fax: +31 24 3540103

E-mail: <u>joostphdrenth@cs.com</u>